

Citation: Hui D, Mehrabi S, Quimby AE, Chen T, Chen S, Park J, et al. (2023) Gene burden analysis identifies genes associated with increased risk and severity of adult-onset hearing loss in a diverse hospital-based cohort. PLoS Genet 19(1): e1010584. https://doi.org/10.1371/journal. pgen.1010584

Editor: Rick Friedman, University of California San Diego, UNITED STATES

Received: September 1, 2022

Accepted: December 20, 2022

Published: January 19, 2023

Copyright: © 2023 Hui et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Summary statistics for individual gene-phenotype associations are provided in S3 and S6 Tables. Summary statistics for the genome-wide association study are available on the EBI GWAS catalog (https://www. ebi.ac.uk/gwas/) under accession GCST90244103. Summary data for genetic variants captured using WES in the PMBB can be accessed via the PMBB Genome Browser (https://pmbb.med.upenn.edu/ allele-frequency/). Individual-level data are not publicly available due to research participant RESEARCH ARTICLE

Gene burden analysis identifies genes associated with increased risk and severity of adult-onset hearing loss in a diverse hospitalbased cohort

Daniel Hui¹, Shadi Mehrabi², Alexandra E. Quimby³, Tingfang Chen¹, Sixing Chen¹, Joseph Park¹, Binglan Li⁴, Regeneron Genetics Center⁵, Penn Medicine Biobank^{6‡}, Michael J. Ruckenstein³, Daniel J. Rader^{1,6,7}, Marylyn D. Ritchie^{1,8}, Jason A. Brant^{3,9}, Douglas J. Epstein^{1®}*, Iain Mathieson^{1®}*

1 Department of Genetics, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 2 Department of Otolaryngology, University of Michigan, Ann Arbor, Michigan, United States of America, 3 Department of Otolaryngology–Head and Neck Surgery, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 4 Department of Biomedical Data Science, Stanford University, Stanford, California, United States of America, 5 Regeneron Genetics Center, Tarrytown, New York, United States of America, 6 Institute for Translational Medicine and Therapeutics, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 7 Department of Medicine, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 8 Institute for Biomedical Informatics, Perelman School of Medicine, University of Pennsylvania, Philadelphia, Pennsylvania, United States of America, 9 Department of Otolaryngology–Head and Neck Surgery, Corporal Michael J. Crescenz VAMC, Philadelphia, Pennsylvania, United States of America

So These authors contributed equally to this work.

‡ A full list of contributors is available in <u>S1 Note</u>.

* epsteind@pennmedicine.upenn.edu (DJE); mathi@pennmedicine.upenn.edu (IM)

Abstract

Loss or absence of hearing is common at both extremes of human lifespan, in the forms of congenital deafness and age-related hearing loss. While these are often studied separately, there is increasing evidence that their genetic basis is at least partially overlapping. In particular, both common and rare variants in genes associated with monogenic forms of hearing loss also contribute to the more polygenic basis of age-related hearing loss. Here, we directly test this model in the Penn Medicine BioBank-a healthcare system cohort of around 40,000 individuals with linked genetic and electronic health record data. We show that increased burden of predicted deleterious variants in Mendelian hearing loss genes is associated with increased risk and severity of adult-onset hearing loss. As a specific example, we identify one gene-TCOF1, responsible for a syndromic form of congenital hearing lossin which deleterious variants are also associated with adult-onset hearing loss. We also identify four additional novel candidate genes (COL5A1, HMMR, RAPGEF3, and NNT) in which rare variant burden may be associated with hearing loss. Our results confirm that rare variants in Mendelian hearing loss genes contribute to polygenic risk of hearing loss, and emphasize the utility of healthcare system cohorts to study common complex traits and diseases.

privacy concerns; however, requests from accredited researchers for access to additional data relevant to this manuscript can be made by contacting the Penn Biobank at biobank@upenn. edu.

Funding: This study was funded by a Translational Genetics Initiative grant from the University of Pennsylvania Health System (J.A.B, D.J.E.), the Boucai Innovation Fund in Auditory Genomics (D.J. E.) and the National Institutes of Health R01 DC006254 (D.J.E) and R35 GM133708 (I.M.). The Penn Medicine BioBank is supported by the Perelman School of Medicine at University of Pennsylvania, a gift from the Smilow family, and the National Center for Advancing Translational Sciences of the National Institutes of Health under CTSA award number UL1TR001878. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health or other funders. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: Regeneron Genetics Center is part of Regeneron Pharmaceuticals. The other authors have declared that no competing interests exist.

Author summary

Age-related hearing loss is relatively common and has a genetic component to risk, though little is known about specific genes that are involved. Here, we study participants in the Penn Medicine BioBank to identify factors that increase risk of hearing loss. We show that loss-of-function variants in genes that are known to cause Mendelian forms of hearing loss also increase risk of age-related hearing loss. Because many such genes have been identified, they can be used to interpret the results of genome-wide association studies and to investigate the biological basis of age-related hearing loss. Our results also emphasize that many reported Mendelian hearing loss variants are incompletely penetrant and may act cumulatively. Finally, our study shows how hospital-recruited biobank cohorts can aid in the study of common conditions like hearing loss, even when those conditions are not the primary reason for contact with the health system.

Introduction

Hearing loss (HL) is one of the most common age-related conditions. Around 50% of American adults experience some difficulty hearing and 50% of those aged 70 and over have hearing loss to a level that is disabling [1]. Estimates of the heritability of age-related hearing loss range from 30% to 70% [2] and large genome-wide association studies (GWAS) have identified 89 independent loci that are significantly associated with risk of age-related HL though, as with most GWAS results, the effects are small [3-8]. In parallel, extensive work with family-based and exome sequencing studies of patients with congenital hearing loss has identified 124 genes associated with nonsyndromic HL, and 45 associated with syndromic HL [9], although variants in syndromic HL genes can also cause nonsyndromic HL [10]. These variants are usually assumed to act in a Mendelian fashion and are classified as autosomal dominant (DFNA), autosomal recessive (DFNB) or X-linked (DFNX). Prelingual HL is most commonly associated with homozygous variants in DFNB genes whereas heterozygous variants in DFNA genes tend to cause postlingual HL [11,12]. Although age-related and congenital HL are often studied separately, it is clear that there is substantial overlap between genes associated with the two conditions. Heritability and common variant GWAS associations with age-related HL are enriched near congenital HL genes [7,13], and around 25% of loci identified in GWAS for age-related HL overlap with congenital HL genes [3]. Recently, several studies of age-related HL found high burdens of rare or predicted deleterious variants in known HL genes [14-16], an observation that suggests that deleterious variants, or combinations of deleterious variants, in congenital HL genes might contribute to increased risk of age-related HL.

Unlike children, adult patients are rarely assessed for genetic etiology when presenting with hearing loss. As current clinical management of adult-onset HL would not be significantly impacted by results of genetic testing, such investigation would have low immediate clinical utility. However, understanding the genetic basis of age-related HL may have longer-term benefits for translational therapy, particularly if such therapy can be linked to congenital HL genes, many of which are well-studied in mouse or other models. Greater understanding of HL genetics would also enable screening of high risk individuals at an earlier age who might benefit from preventive or restorative treatment options. Because age-related HL is common, many participants in hospital or healthcare system biobank cohorts are likely to have HL (even if it is not the primary reason for their contact with the health system) providing an opportunity to study the genetic basis of age-related HL in large cohorts without specific recruitment.



Fig 1. Project workflow, describing data pre-processing such as inclusion/exclusion criteria for individuals and variants, and final datasets used for analyses. Discovery analyses were performed in PMBB for both common and rare variants. Select results were replicated in UK BioBank and using the results of Praveen et al [8].

https://doi.org/10.1371/journal.pgen.1010584.g001

In this study, we use the Penn Medicine BioBank (PMBB) to investigate the extent to which deleterious variants in known congenital HL genes contribute to polygenic risk of age-related HL and attempt to identify specific genes that contribute to the phenotype (Fig 1).

Results

Hearing loss in the Penn Medicine BioBank cohort

PMBB is a database of linked electronic health records (EHR), biospecimens and genetic data of participants recruited through the Penn Medicine health system [17]. The data analyzed here consist of genetic and EHR data from 40,627 individuals. The cohort is 51% male and 49% female, with a median age of 58 years (S1 Table). Based on comparison with genetic reference populations, approximately 75% of the cohort is of European ancestry and 25% is of African American ancestry. A previous analysis of a subset of 16,657 individuals identified an excess of predicted deleterious variants in DFNA genes in participants with audiometrically-defined HL [16].

We first categorized cases and controls following a common scheme in EHR studies. Phenotypes are defined in terms of phecodes–a standardized hierarchical classification of diseases and traits [18,19]. If a patient has two or more instances of a specific phecode in their records, they are a case. If they have zero instances they are a control and if they have one instance they are missing (NA). This approach yielded 3,304 potential cases and 34,704 potential controls. As mild to moderate hearing loss may not always be accurately reflected in EHR data, we cross-checked using a subset of 1,917 individuals for whom we had audiometric data in the form of audiograms, which are measurements of the quietest sound that an individual can hear at a range of frequencies. We find that, of participants with audiograms, 65% of phecodedefined cases and 27% of phecode-defined controls have audiogram-defined hearing loss based on a conventional Pure Tone Average (PTA) threshold of >25 (S2 Table). Due to this relatively high rate of case/control misclassification, we adopted a hybrid strategy to maximize accurate identification of cases and number of controls. We performed all association tests on degree of hearing loss (degree HL), defined as a continuous variable with values from 0–4. We assigned degree HL to individuals with audiograms based on PTA ranges (degree HL 0–4 corresponding to PTA 0–15, 16–25, 26–40, 41–55, and 56+). For individuals without audiograms, we assigned degree HL 0 to phecode-defined controls and removed phecode cases and NA. This assignment is conservative in the sense that the true mean degree HL of phecode controls is greater than 0, but we assumed that it is 0 and would therefore tend to underestimate any associations.

Deleterious variant burden in known HL genes is associated with risk and degree of HL

Next, we investigated the frequency of predicted deleterious variants in 173 known HL genes (Fig 2, S2 and S3 Tables). We found that 72.8% of 35,397 "controls" (HL 0–1 or Phecode 0) and 74.0% of 1,110 "cases" (HL 2–4) carried at least one deleterious variant in a known HL gene (Fisher's exact test p = 0.51). Similarly, 6.80% of controls and 7.93% of cases (p = 0.147) carry at least one variant in a known HL gene annotated as "pathogenic" or "likely pathogenic"



Fig 2. Percentage of hearing loss (HL) cases and controls with at least one variant in different gene and variant categories. "All" refers to the full list of known HL genes, "DFNB" and "DFNA" percentages are only for genes with recessive and dominant forms of Mendelian inheritance, respectively, "audiogram only" percentages are restricted to only individuals with audiograms, and percentages with "ClinVar variants" only included variants with pathogenic/likely pathogenic ClinVar annotations. P-values from Fisher's exact test of case versus control carrier counts.

	Beta	SE	Р	
All variants	0.00483	0.00224	0.0308	
ClinVar variants	0.0179	0.00975	0.0660	
Non-ClinVar variants	0.00467	0.00228	0.0403	
Variants in DFNA genes	0.00701	0.00344	0.0415	
Variants in DFNB genes	0.00269	0.00314	0.392	
Variants in DFNA+B genes	0.00687	0.00820	0.403	
÷				

Table 1. Association of total burden in known hearing	ng loss genes in PMBB with degree HI
---	--------------------------------------

https://doi.org/10.1371/journal.pgen.1010584.t001

in ClinVar [20], a database of genetic variants reported to be associated with disease. We regressed degree HL on the burden count, defined as the total number of predicted deleterious variants in known HL genes including sex, age, age² and 20 genomic principal components as covariates, and find that degree HL is associated with burden count ($\beta = 4.8 \times 10^{-3}$ per-variant, p = 0.03, Table 1). Restricting to subsets of variants we find larger point effects for pathogenic or likely pathogenic ClinVar variants compared to other variants, but the effects of variants in DFNA and DFNB genes were not significantly different. The association is still significant and ten-fold larger in magnitude if we restrict to N = 1,917 individuals with audiograms ($\beta = 4.7 \times 10^{-2}$ per-variant, p = 0.04). Next, we estimated the effect of burden for each degree HL independently, confirming that gene burden is associated with both the presence and severity of hearing loss (Fig 3). Finally, we replicated this result in UK BioBank (UKB) using binary hearing loss as a phenotype since audiograms were not available ($p = 1.84 \times 10^{-12}$ for association between loss-of-function variant burden in known HL genes and HL). Though absolute effect sizes are not directly comparable between UKB and PMBB because of the different phenotypes, we noted that in UKB, as in PMBB, the effect of each ClinVar variant was four times





	Beta	SE	Р
All variants	0.041	0.0059	1.84×10 ⁻¹²
ClinVar variants	0.141	0.0248	1.32×10 ⁻⁰⁸
Non-ClinVar variants	0.036	0.0060	4.11×10 ⁻⁰⁹
Variants in DFNA genes	0.039	0.0094	3.19×10 ⁻⁰⁵
Variants in DFNB genes	0.034	0.0081	2.40×10 ⁻⁰⁵
Variants in DFNA+B genes	0.098	0.0204	1.79×10^{-06}

Table 2.	Association of	of total b	urden in l	known	hearing	loss	genes v	with	HL ir	n UKB
----------	----------------	------------	------------	-------	---------	------	---------	------	-------	-------

https://doi.org/10.1371/journal.pgen.1010584.t002

the size of each non-ClinVar variant, the effects of variants in DFNA and DFNB genes were similar, and the effects of variants in genes that can act as both DFNA and DFNB were larger (Table 2). Taken together, these results demonstrate that increased burden of known and predicted deleterious variants in Mendelian HL genes is associated with increased risk and severity of adult-onset HL.

Deleterious variants in TCOF1 are associated with age related HL

We tested for association between predicted deleterious variant burden of individual HL genes, and degree HL, using all individuals (<u>S3 Table</u>). We found one significant gene, *TCOF1*, at Benjamini-Hochberg false discovery rate (FDR) < 0.05 (FDR = 7.2×10^{-4} , p = 5.2×10^{-6}) (Fig 4, Table 3). We replicated this association in a previously reported case-control study of self-



Fig 4. QQ-plot for association of deleterious variant burden with degree HL in 173 known HL genes (red, case carriers > 0) and across 373 genes not previously associated with hearing loss (blue, case carriers > 25). Genes significant at FDR<0.05 are labeled. Genomic inflation factor $\lambda = 0.78$ for known HL genes and 1.44 for genes not previously associated.

Gene	Beta	SE	Р	Carriers	Carrier Freq.	FDR
TCOF1	0.798	0.175	5.20×10 ⁻⁶	8	0.00022	0.0008788
ESRRB	0.148	0.0434	7.00×10^{-4}	129	0.0035	0.05915

Table 3. Known hearing loss gene burden associated with degree hearing loss (FDR < 0.1) in PMBB.

https://doi.org/10.1371/journal.pgen.1010584.t003

reported HL by Praveen et al. which performed gene burden analysis in approximately 108,000 cases and 330,000 controls [8] (minimum $p = 8.0 \times 10^{-4}$ over 24 burden models, S4 Table). The next most significant gene in our data was *ESRRB* (FDR = 0.06, p = 7.0×10^{-4}), a nuclear receptor associated with autosomal recessive hearing loss [21]. However, this association did not replicate (minimum p = 0.011 over 24 burden models, S4 Table).

Autosomal dominant mutations in *TCOF1* cause Treacher Collins Syndrome, which involves craniofacial deformities and conductive hearing loss due to abnormal neural crest cell development [22,23]. The majority of Treacher Collins Syndrome cases are caused by truncating mutations that result in *TCOF1* haploinsufficiency. We identified 8 carriers of predicted loss-of-function or pathogenic missense variants in our dataset (S5 Table). None of these variants were previously reported in ClinVar. Of these carriers, 2 had moderate to severe hearing loss with degrees 3 and 4, but only one (the individual with HL 4) had a clinical diagnosis of Treacher Collins Syndrome. After manually reviewing charts of all carriers, we found that an additional three individuals had reports of hearing loss, and one other individual reported tinnitus. Therefore, 4/7 carriers without a clinical diagnosis of Treacher Collins syndrome had some evidence of hearing impairment. None of these individuals' charts reported evidence of craniofacial deformities or other Treacher Collins symptoms. These results demonstrate that damaging variants in *TCOF1* that do not manifest in Treacher Collins Syndrome may none-theless increase risk or severity of hearing loss.

Four candidate novel HL-associated genes

We next searched for genes contributing to adult-onset HL but are not known to cause congenital HL–i.e., were not on our list of 173 known HL genes. We tested for association of individual gene burden with degree of hearing loss, using the same model as for the total gene burden in previous sections. In initial exome-wide results, we observed extreme inflation in test statistics. Randomizing the phenotype did not reduce this inflation indicating that it was due to poor calibration of test statistics rather than uncorrected population stratification [e.g., 17,24]. We therefore filtered genes using a threshold of >25 case carriers, which restricted analyses to a set of 373 genes, and produced well-calibrated test-statistics.

Of these 373 genes, four (*COL5A1*, *HMMR*, *RAPGEF3*, and *NNT*) were significant at FDR < 0.05 (Tables 4 and S6, Fig 4). None of these genes were significantly associated with binary HL in the Praveen et al. study [8] (S4 Table). Nonetheless, all four genes are expressed in cell types in the mouse cochlea that are relevant for hearing (Fig 5). Moreover, several of these genes have additional evidence in humans or mice that supports a role in hearing loss. In particular variants in *COL5A1*, which encodes a subunit of type V collagen, are associated with

Table 4.	Novel	gene burden	associated	with degr	ee hearing	loss (F	DR <	0.05) in	a PMBB.

Gene	Beta	SE	р	Carriers	Carrier Freq.	FDR
COL5A1	0.0586	0.0141	3.31×10 ⁻⁵	1255	0.034	0.0123
HMMR	0.0974	0.0245	6.89×10 ⁻⁵	407	0.011	0.0128
RAPGEF3	0.0731	0.0196	1.88×10^{-4}	647	0.018	0.0189
NNT	0.0507	0.0136	2.03×10^{-4}	1013	0.028	0.0189



Fig 5. Expression of genes identified in our study in specific cell types from a representative transverse section through the mouse cochlea. The heat map was generated from previously published bulk and single cell RNA-seq datasets deposited in the gEAR portal (https://umgear.org/). Abbreviations: Deiters' cells (DC), hair cells (HC), inner hair cell (IHC), outer hair cells (OHCs), pillar cells (PC), Reissner's membrane (RM), spiral ganglion neurons (SGN) Type 1 (T1) and Type 2 (T2), spiral ligament (SL), spindle-root cells (SR), stria vascularis (SV), support cells (SC).

https://doi.org/10.1371/journal.pgen.1010584.g005

Ehlers-Danlos syndrome, a connective tissue disorder that may also involve the auditory system, including conductive and sensorineural hearing loss [25–27]. *RAPGEF3/EPAC1* is a cAMP sensitive guanine nucleotide exchange factor for the small GTPases RAP1 and RAP2.

Rapgef3/Epac1 knockout mice display pancreatic beta-cell dysfunction and metabolic syndrome [28], which are known risk factors for sensorineural hearing loss [29]. *Rapgef3/Epac1* is also upregulated in response to noise in rats and its pharmacological inhibition has been shown to attenuate inner ear pathology caused by noise exposure [30]. The *nicotinamide nucleotide transydrogenase* (*NNT*) gene encodes an integral protein of the inner mitochondrial membrane. Mice with *Nnt* mutations exhibit impaired insulin secretion, which is also known to increase risk of hearing loss [31,32]. Thus, plausible causal mechanisms may be inferred for 3 out of 4 candidate novel genes associated with adult-onset HL that were identified in our rare variant gene burden analysis.

Genome-wide association analysis identifies *PLPPR5* as a single candidate locus

We conducted common variant GWAS under an additive model in all study individuals, for variants with a minor allele frequency greater than 1%. As with the burden analysis, we randomized the phenotype and observed quantile-quantile plots (QQ-plots) of the randomized and observed results, and increased the allele frequency cutoff until inflation was not observed in the randomized results (genomic inflation constant $\lambda = 1.01$). We identified a single genome-wide significant locus upstream of PLPPR5 (lead SNP chr1:99058420:C:T, MAF = 0.012, $p = 8.27 \times 10^{-9}$, Fig 6). *PLPPR5* has previously not been associated with hearing loss-nevertheless, single cell RNA-seq analysis of mouse cochlea does indicate selective Plppr5 expression in inner hair cells (Fig 5). Mouse knockout of *Plppr5* does not result in HL [33], although hearing was only measured in young mice, so an age-related HL phenotype would not be revealed in this analysis. This association did not replicate in the Praveen et al. GWAS results (p = 0.759) [8] and would need to be replicated elsewhere before being considered reliable. Conversely, of the 53 genome-wide significant lead variants reported in that study, 45 were present in our dataset of which 9 had p-values <0.05, significantly more than expected by chance (binomial p-value 0.0003, S7 Table). We also replicated at p<0.05 2 out of 12 novel rare variants identified by Ivarsdottir et al. [5] that were also present in our dataset (S8 Table), which is not significantly more than expected by chance (p = 0.11) but unsurprising since we have low power for rare variant associations.

Polygenic score and rare variant burden score have low predictive power

Finally, we computed polygenic risk scores (PRS) using PRS-CS [34] and a previously reported GWAS of hearing loss in UK BioBank [35]. The PRS is constructed by summing up all the effects of all the GWAS-identified variants carried by an individual and provides an estimate of their genetic risk. We summarize the predictive power of this score using the incremental R^2 –the difference in R^2 between a linear model of degree HL with and without the PRS (but including other covariates, Methods). The incremental R^2 of the PRS is 0.19% (bootstrap



Fig 6. Local association plot of GWAS results at the PLPPR5 locus (generated with LocusZoom [48]).

	All individuals			Audiogram only		
Ancestry	All	EUR	AFR	All	EUR	AFR
PRS	0.153%	0.186%	0.032%	0.970%	1.00%	0.236%
HL burden	0.012%	0.017%	0.003%	0.166%	0.274%	0.000%

Table 5. Tredictive power (incrementar K) of polygeme risk scores and burden of detectious variants in known the genes for degree the in the	x (incremental R^2) of polygenic risk scores and burden of deleterious variants in known HL genes for degree HL in PMB
--	---

https://doi.org/10.1371/journal.pgen.1010584.t005

standard error 0.054%), rising to 1% if we restrict to individuals with audiogram data (Table 5). As observed for other traits [36], the predictive power of the PRS is much lower in individuals of African American ancestry, compared to those of European ancestry. The predictive power of the rare variant burden score is much lower than the PRS (Table 5), and also lower in African American compared to European ancestry individuals. This is surprising because, unlike the PRS, the burden score was not directly trained in a European ancestry cohort. Despite these low incremental R^2 , both scores have the potential to identify a subset of individuals at high risk of HL. For example, individuals in both the top 10% of the PRS distribution and the top 10% of HL gene burden distribution had an odds ratio of 2.9 (for degree HL 2 or greater), similar to individuals in the top 1% of the PRS distribution (odds ratio 2.6).

Discussion

Our study illustrates both the opportunities and challenges of using EHR data to study a common complex condition like hearing loss. In particular, the subset of individuals in our data with audiograms makes it clear that EHR-derived phenotypes are highly inaccurate for this phenotype, perhaps because phecodes are largely constructed using hospital billing codes and hearing loss is not typically the primary reason for hospital visits. Published large GWAS studies have relied largely on self-reported hearing loss in UK Biobank [3,5] which may be similarly inaccurate, though some studies include a subset of individuals with audiograms. From an analysis perspective, it may make more sense to think of audiogram measurements, selfreported HL and EHR-derived HL as different, though related, phenotypes. For example, our stricter "case" definition entails a case proportion of 3.0% in PMBB, compared to 26.4% in UK Biobank. Because we had audiometric data only for a relatively small proportion of our cohort, we actually used a hybrid strategy where audiogram data identified "cases" and people with neither audiograms nor EHR-derived HL were assumed not to have hearing loss (i.e., "controls", HL 0). We chose this strategy because it maximizes sample size while being conservative, in the sense that it would tend to underestimate associations with HL. Individuals without EHR-derived HL but with audiograms have a mean HL of 0.8 on our 0-4 scale, providing one estimate of the true baseline level in these individuals and suggesting that our effect sizes estimates would be biased downwards. On the other hand, this could be an overestimate relative to the general population since it is possible that people who have audiograms are enriched for people with some hearing loss. Ultimately, the limited number of individuals with audiometric data is probably the major factor limiting the power of our study. One strategy that other studies have used is to analyze audiogram and self-reported data separately and to then meta-analyze the results [5]. It remains unclear what the best way to combine these phenotypes is. This heterogeneity in phenotype may contribute to the lack of replication of some of our findings in other datasets and collecting larger datasets of genotyped individuals with audiometric data should be a priority for further research.

We identified and replicated a novel association of *TCOF1* variants with adult-onset HL. While *TCOF1* is known to cause congenital HL as part of Treacher-Collins syndrome, it has not previously been reported to increase risk of adult-onset HL. While hearing loss in Treacher-Collins syndrome is generally conductive due to malformation of middle ear structures, in at least one of our *TCOF1* cases hearing loss is sensorineural, suggesting a different mechanism. This result is an example of how knowledge about genes associated with earlyonset or developmental conditions can be used to prioritize gene discovery for related adultonset conditions. Similar observations have been made about the excess contribution of genes that are essential or involved in developmental disorders to psychiatric traits such as autism [37] and schizophrenia [38].

Our observation that rare variant burden in known HL genes is associated with both risk and severity of adult onset HL supports previous observations that common variant heritability is enriched around known HL genes [13] and that deleterious variants in congenital HL genes are associated with age-related HL [14–16]. Conversely, the high rate of predicted deleterious coding mutations in controls emphasizes that many of these variants have low penetrance [14], and their effects on age-related HL are cumulative and polygenic, rather than Mendelian. Further, the low predictive power of both rare variant burden and polygenic scores suggests that these are not yet particularly useful tools for HL risk prediction. One limitation of our analysis is that we cannot distinguish the extent to which the polygenic contribution to disease from congenital HL genes is uniformly distributed across those genes, as opposed to being concentrated in a relatively small number of genes which we lack power to identify individually.

We also identified five novel genes potentially associated with HL-four through gene burden analysis and one through GWAS. Although these genes are all biologically plausible, these associations are only modestly supported and none of them replicated in the Praveen et al. study. We therefore consider these associations as only candidates until replicated. One limitation is that Praveen et al. used self-reported hearing loss as opposed to audiometric data. Another is that for rare variants we only looked at coding variation and not regulatory variation at these genes, which might explain a larger proportion of heritability.

Another opportunity and challenge of our dataset is the level of ancestral diversity. Approximately 25% of the individuals in the Penn Medicine BioBank are of African American ancestry. Performing genetic discovery within and across cohorts of diverse ancestry is both a key goal and largely open question in human genetic research today [39,40]. While previous studies have meta-analyzed ancestry-specific results [17], we instead analyzed all individuals together, correcting for ancestry using genome-wide principal components. Exome-based studies may in fact be more amenable to cross-ancestry analysis since they assume that the loss-of-function variants that are counted in the test are in fact the causal variants. Therefore, unlike GWAS these tests should not be affected by differences in linkage disequilibrium and tagging of causal variants across populations. We expected that the effect of coding variants would be similar across ancestries. However, the observation that the predictive power of rare variant burden on HL is lower in African American ancestry individuals compared to European ancestry individuals (like polygenic scores [36]) calls this expectation into question and requires further investigation.

In summary, we used biobank data and a prioritized list of known HL genes to identify novel associations with adult-onset HL. We showed that burden of deleterious variants in known HL genes is robustly associated with HL and identified five additional candidate genes through exome- and genome-wide analysis. Because hearing loss is so common in older individuals, healthcare cohorts like the Penn Medicine Biobank have great potential to allow gene discovery, and validation of predictive tests, even if not specifically collected for that purpose, and we expect that these approaches will become increasingly common as such cohorts expand in size.

Methods

Ethics statement

The collection, storage and analysis of biospecimens, genetic data and data derived from electronic health records as part of the Penn Medicine BioBank is approved under University of Pennsylvania IRB protocol #813913. Each participant provided written informed consent.

Study setting and participants

The Penn Medicine BioBank (PMBB) has recruited approximately 60,000 participants and hosts data from patients of clinical practice sites of the University of Pennsylvania Health System. Each participant provided informed consent regarding storage of biological specimens, genetic sequencing, and access to all available EHR data. This study was approved by the Institutional Review Board of the University of Pennsylvania.

Data collection

Exome sequences were generated by the Regeneron Genetics Center (Tarrytown, NY). Sequencing was performed in 2020 with the custom IDT xGen v1 exome capture platform and sequenced on an Illumina NovaSeq 6000 system using S4 flow cells. Sequences were mapped and variants were annotated using ANNOVAR, gnomAD, and REVEL (Rare Exome Variant Ensemble Learner), and samples with low exome sequencing coverage (less than 85% of targeted sites at 20X coverage), apparent contamination (D-statistic > 0.4), discordance with reported sex, or that were duplicates, were removed, as previously described [17,41]. After variant calling using the weCall variant caller, we set any site with fewer than seven reads to missing and required variants to have at least one homozygote call or at least one heterozygote call supported by at least 15% of reads. We inferred ancestry by projecting array genotype data onto principal component axes defined by individuals from the 1000 Genomes Project [42] and fitting a Gaussian mixture model. We also removed individuals such that all were unrelated up to the 3rd degree. A total of 9,356 individuals considered African American ancestry and 27,151 individuals considered European ancestry were included after quality control measures including removal of individuals with missing covariates.

Phenotyping

We extracted audiometric data from a clinical database (AudBase) of audiograms performed at the Hospital of the University of Pennsylvania audiology practice between May 5, 2013 and February 25, 2021–1,917 audiograms were available for the 36,507 study individuals. From the clinical database, we calculated pure-tone average (PTA) for bone and air conduction using the arithmetic mean of the hearing threshold in decibels at 500, 1000, and 2000 Hertz. We recorded air conduction PTA of the worse ear and assigned HL levels according to previously published categorization: PTA 0–15 (degree HL 0), PTA 16–25 (degree HL 1, mild), PTA 26–40 (degree HL 2, moderate), PTA 41–55 (degree HL 3, severe), and PTA 56+ (degree HL 4, profound) [43]. To identify additional controls phecode 389 (Hearing Loss) was used, counting individuals as controls if they had zero EHR records of this phecode. In total, we included 35,397 controls and 1,110 hearing loss cases in analyses (with degree HL 2, N = 334 degree HL 3, and N = 260 degree HL 4).

Rare variant gene burden

To select variants to include in gene burden analysis, we used a combination of minor allele frequency filters and predicted pathogenicity. First, we removed variants if their gnomAD allele frequency in African and Non-Finnish European individuals was greater than .001. We then filtered remaining variants on an allele frequency of .01 in the entire dataset. Next, we restricted to predicted loss-of-function (pLoF) variants, namely frameshift mutations, stopgains, and those that disrupt canonical splice sites. We also included nonsynonymous missense variants after filtering to restrict to those with high predicted pathogenicity, defined as the Rare Exome Variant Ensemble Learner (REVEL) score being greater than 0.6 [41]. We annotated variants based on their ClinVar annotation, considering known pathogenic variants to be those with annotation "Pathogenic", "Likely pathogenic", or "Pathogenic/Likely pathogenic". Ultimately, we identified 11,641 predicted deleterious variants, 575 of which were reported pathogenic variants. For known HL genes, we included 173 autosomal genes, of which 66 were DFNA, 97 DFNB, and 10 had multiple inheritance patterns. For the gene burden analyses, any included variants in a gene were summed to create a single gene burden for that individual using BioBin [44], which we used in downstream association analyses. For the total gene burden analyses in known HL genes, we summed together all such variants in known HL genes for each individual. We tested for association using linear regression with degree HL as the outcome and including age, age², sex, and genome-wide PCs 1–20 as covariates.

We performed permutation analyses to determine the (case) carrier cut-off and minor allele frequency thresholds for the burden tests, as we observed test statistic inflation in the QQ-plots of the $-\log_{10}(p)$ from the association tests without additional filtering. Briefly, we randomly permuted the phenotype and re-ran the association tests. For known HL genes, we assumed that variants in these genes would be deleterious, as deleterious variants in these genes were how they were discovered and thus we filtered based on case carriers >0 instead of thresholding based on carriers only. When expanding analyses to all genes, we again used a case carrier threshold, but were much more conservative as we were attempting novel discovery-we increased the case carrier threshold in increments of 5 until we observed very minimal test statistic inflation in the randomly permuted $-\log_{10}(p)$ QQ-plot, and chose a threshold of >25 case carriers. To correct for multiple comparisons, we treated burden in known and novel HL genes separately, and used Benjamini-Hochberg correction.

Common variants

Array genotypes were processed according to these specifications (https://pmbb.med.upenn. edu/data-access/data-release/2020_2_0/imputed/), and were imputed to the TOPMed reference panel [45] using the Michigan Imputation Server [46]. Common variant analysis was performed on the same set of individuals that were used for the rare variant burden analysis. We filtered variants on imputation $R^2 > .30$ and used the same modeling setup and covariates as with the burden tests. We performed association tests using Plink 1.9 [47]. To determine the minor allele frequency threshold, we ran similar permutation analyses for GWAS of common variants as we did with the burden analyses. After randomly permuting the phenotype, we started at a minor allele frequency threshold of .1%, increased it to .5%, and observed minimal test statistic inflation at a threshold of 1%.

Polygenic risk scores

We calculated PRS for PMBB individuals with PRS-CS [34] using default settings, including restricting SNPs to a set of 1,009,605 HapMap3 SNPs. We used European ancestry GWAS

summary statistics from UKB [35]. We report incremental R^2 , calculated by subtracting the difference in R^2 between HL regressed on all covariates (age, age², sex, PCs 1–20) and PRS from the R^2 when regressing HL just on the covariates.

Replication

We replicated the association between gene burden and HL in UK Biobank. Hearing loss cases were defined as those who answered 'Yes' for both "Hearing difficulty/problems" and "Hearing difficulty/problems with background noise", and controls were individuals who answered 'No' for both-remaining individuals were removed. We note that this phenotyping definition is similar to previous analyses of hearing loss in UKB [3]. 148,970 (39,272 cases and 109,698 controls (26.4% cases)) European ancestry participants with available exome sequencing were included. We defined predicted deleterious variants as above and performed logistic regression including age, age², sex, and the top 5 genetic principal components provided by UKB.

We replicated novel significant gene and GWAS associations using results from Praveen et al. [8]. Briefly, this study reports a meta-analysis of five cohorts of European ancestry, including UK Biobank, with 125,749 cases and 469,497 for common variants and 108,415 cases and 329,581 controls for rare variant burden analyses. Since the phenotype and variant definitions were different between this study and our dataset, we tested 24 different models per gene (6 different allele frequency thresholds and 4 definitions of variant deleteriousness), and considered a gene to replicate if the minimum P-value was below a Bonferroni-corrected significance threshold of 0.05/24.

Supporting information

S1 Table. Basic demographic properties of PMBB cohort. (XLSX)

S2 Table. Comparison between Phecode-defined case/control status and Audiograms. (XLSX)

S3 Table. List of 173 known congenital hearing loss (HL) genes and their Mendelian inheritance pattern(s).

(XLSX)

S4 Table. Replication results for known and novel HL genes identified through LoF burden association.

(XLSX)

S5 Table. LoF and predicted deleterious TCOF1 mutations. (XLSX)

S6 Table. LoF burden association results for 373 genes not previously reported as HL genes with at least 25 carriers. (XLSX)

S7 Table. Attempted replication results for associations reported by Praveen et al. [8]. (XLSX)

S8 Table. Attempted replication results for novel associations reported by Ivarsdottier et al. [5]. (XLSX)

S1 Note. Members of the Penn Medicine BioBank Consortium. (PDF)

Acknowledgments

We acknowledge the Penn Medicine BioBank (PMBB) for providing data and thank the patient-participants of Penn Medicine who consented to participate in this research program. We would also like to thank the Penn Medicine BioBank team and Regeneron Genetics Center for providing genetic variant data for analysis. The UK BioBank resource was used under Application 54239.

Author Contributions

- **Conceptualization:** Daniel Hui, Shadi Mehrabi, Michael J. Ruckenstein, Daniel J. Rader, Marylyn D. Ritchie, Jason A. Brant, Douglas J. Epstein, Iain Mathieson.
- **Data curation:** Daniel Hui, Shadi Mehrabi, Alexandra E. Quimby, Tingfang Chen, Sixing Chen, Joseph Park, Binglan Li, Jason A. Brant.
- Formal analysis: Daniel Hui, Shadi Mehrabi, Alexandra E. Quimby, Tingfang Chen, Sixing Chen, Joseph Park, Binglan Li, Douglas J. Epstein, Iain Mathieson.
- Project administration: Daniel J. Rader, Marylyn D. Ritchie, Douglas J. Epstein.
- Supervision: Michael J. Ruckenstein, Daniel J. Rader, Marylyn D. Ritchie, Jason A. Brant, Douglas J. Epstein, Iain Mathieson.

Writing - original draft: Daniel Hui, Iain Mathieson.

Writing - review & editing: Daniel Hui, Douglas J. Epstein, Iain Mathieson.

References

- National Institute on Deafness and Other Communication Disorders. Quick Statistics About Hearing 2021 [cited 2022 May 11]. Available from: https://www.nidcd.nih.gov/health/statistics/quick-statisticshearing.
- Wells HRR, Newman TA, Williams FMK. Genetics of age-related hearing loss. J Neurosci Res. 2020; 98(9):1698–704. https://doi.org/10.1002/jnr.24549 PMID: 31989664
- Wells HRR, Freidin MB, Zainul Abidin FN, Payton A, Dawes P, Munro KJ, et al. GWAS Identifies 44 Independent Associated Genomic Loci for Self-Reported Adult Hearing Difficulty in UK Biobank. Am J Hum Genet. 2019; 105(4):788–802. https://doi.org/10.1016/j.ajhg.2019.09.008 PMID: 31564434
- Nagtegaal AP, Broer L, Zilhao NR, Jakobsdottir J, Bishop CE, Brumat M, et al. Genome-wide association meta-analysis identifies five novel loci for age-related hearing impairment. Sci Rep. 2019; 9 (1):15192. https://doi.org/10.1038/s41598-019-51630-x PMID: 31645637
- Ivarsdottir EV, Holm H, Benonisdottir S, Olafsdottir T, Sveinbjornsson G, Thorleifsson G, et al. The genetic architecture of age-related hearing impairment revealed by genome-wide association analysis. Commun Biol. 2021; 4(1):706. https://doi.org/10.1038/s42003-021-02224-9 PMID: 34108613
- Hoffmann TJ, Keats BJ, Yoshikawa N, Schaefer C, Risch N, Lustig LR. A Large Genome-Wide Association Study of Age-Related Hearing Impairment Using Electronic Health Records. PLoS Genet. 2016; 12 (10):e1006371. https://doi.org/10.1371/journal.pgen.1006371 PMID: 27764096
- Trpchevska N, Freidin MB, Broer L, Oosterloo BC, Yao S, Zhou Y, et al. Genome-wide association meta-analysis identifies 48 risk variants and highlights the role of the stria vascularis in hearing loss. Am J Hum Genet. 2022. https://doi.org/10.1016/j.ajhg.2022.04.010 PMID: 35580588
- Praveen K, Dobbyn L, Gurski L, Ayer AH, Staples J, Mishra S, et al. Population-scale analysis of common and rare genetic variation associated with hearing loss in adults. Commun Biol. 2022; 5(1):540. https://doi.org/10.1038/s42003-022-03408-7 PMID: 35661827
- 9. Van Camp G, Smith RDH. Hereditary Hearing Loss Homepage 2022 [cited 2022 12 May]. Available from: https://hereditaryhearingloss.org.

- Bademci G, Cengiz FB, Foster Ii J, Duman D, Sennaroglu L, Diaz-Horta O, et al. Variations in Multiple Syndromic Deafness Genes Mimic Non-syndromic Hearing Loss. Sci Rep. 2016; 6:31622. https://doi. org/10.1038/srep31622 PMID: 27562378
- Petit C. Genes responsible for human hereditary deafness: symphony of a thousand. Nat Genet. 1996; 14(4):385–91. https://doi.org/10.1038/ng1296-385 PMID: 8944017
- Marazita ML, Ploughman LM, Rawlings B, Remington E, Arnos KS, Nance WE. Genetic epidemiological studies of early-onset deafness in the U.S. school-age population. Am J Med Genet. 1993; 46 (5):486–91. https://doi.org/10.1002/ajmg.1320460504 PMID: 8322805
- Kalra G, Milon B, Casella AM, Herb BR, Humphries E, Song Y, et al. Biological insights from multi-omic analysis of 31 genomic risk loci for adult hearing difficulty. PLoS Genet. 2020; 16(9):e1009025. https:// doi.org/10.1371/journal.pgen.1009025 PMID: 32986727
- Lewis MA, Nolan LS, Cadge BA, Matthews LJ, Schulte BA, Dubno JR, et al. Whole exome sequencing in adult-onset hearing loss reveals a high load of predicted pathogenic variants in known deafnessassociated genes and identifies new candidate genes. BMC Med Genomics. 2018; 11(1):77. <u>https://doi.org/10.1186/s12920-018-0395-1</u> PMID: 30180840
- Boucher S, Tai FWJ, Delmaghani S, Lelli A, Singh-Estivalet A, Dupont T, et al. Ultrarare heterozygous pathogenic variants of genes causing dominant forms of early-onset deafness underlie severe presbycusis. Proc Natl Acad Sci U S A. 2020; 117(49):31278–89. <u>https://doi.org/10.1073/pnas.2010782117</u> PMID: 33229591
- Ahmadmehrabi S, Li B, Hui D, Park J, Ritchie M, Rader DJ, et al. A Genome-First Approach to Rare Variants in Dominant Postlingual Hearing Loss Genes in a Large Adult Population. Otolaryngol Head Neck Surg. 2022; 166(4):746–52. https://doi.org/10.1177/01945998211029544 PMID: 34281439
- Park J, Lucas AM, Zhang X, Chaudhary K, Cho JH, Nadkarni G, et al. Exome-wide evaluation of rare coding variants using electronic health records identifies new gene-phenotype associations. Nat Med. 2021; 27(1):66–72. https://doi.org/10.1038/s41591-020-1133-8 PMID: 33432171
- Denny JC, Bastarache L, Ritchie MD, Carroll RJ, Zink R, Mosley JD, et al. Systematic comparison of phenome-wide association study of electronic medical record data and genome-wide association study data. Nat Biotechnol. 2013; 31(12):1102–10. https://doi.org/10.1038/nbt.2749 PMID: 24270849
- Bastarache L. Using Phecodes for Research with the Electronic Health Record: From PheWAS to PheRS. Annu Rev Biomed Data Sci. 2021; 4:1–19. https://doi.org/10.1146/annurev-biodatasci-122320-112352 PMID: 34465180
- Landrum MJ, Lee JM, Benson M, Brown GR, Chao C, Chitipiralla S, et al. ClinVar: improving access to variant interpretations and supporting evidence. Nucleic Acids Res. 2018; 46(D1):D1062–D7. <u>https://</u> doi.org/10.1093/nar/gkx1153 PMID: 29165669
- Collin RW, Kalay E, Tariq M, Peters T, van der Zwaag B, Venselaar H, et al. Mutations of *ESRRB* encoding estrogen-related receptor beta cause autosomal-recessive nonsyndromic hearing impairment DFNB35. Am J Hum Genet. 2008; 82(1):125–38. https://doi.org/10.1016/j.ajhg.2007.09.008 PMID: 18179891
- Wise CA, Chiang LC, Paznekas WA, Sharma M, Musy MM, Ashley JA, et al. *TCOF1* gene encodes a putative nucleolar phosphoprotein that exhibits mutations in Treacher Collins Syndrome throughout its coding region. Proc Natl Acad Sci U S A. 1997; 94(7):3110–5. <u>https://doi.org/10.1073/pnas.94.7.3110</u> PMID: 9096354
- 23. Katsanis SH, Jabs EW. Treacher Collins Syndrome. In: Adam MP, Ardinger HH, Pagon RA, Wallace SE, Bean LJH, Gripp KW, et al., editors. GeneReviews((R)). Seattle (WA)1993.
- 24. Cirulli ET, White S, Read RW, Elhanan G, Metcalf WJ, Tanudjaja F, et al. Genome-wide rare variant analysis for thousands of phenotypes in over 70,000 exomes from two cohorts. Nat Commun. 2020; 11 (1):542. https://doi.org/10.1038/s41467-020-14288-y PMID: 31992710
- De Paepe A, Nuytinck L, Hausser I, Anton-Lamprecht I, Naeyaert JM. Mutations in the COL5A1 gene are causal in the Ehlers-Danlos syndromes I and II. Am J Hum Genet. 1997; 60(3):547–54. PMID: 9042913
- 26. Ritelli M, Dordoni C, Venturini M, Chiarelli N, Quinzani S, Traversa M, et al. Clinical and molecular characterization of 40 patients with classic Ehlers-Danlos syndrome: identification of 18 COL5A1 and 2 COL5A2 novel mutations. Orphanet J Rare Dis. 2013; 8:58. https://doi.org/10.1186/1750-1172-8-58 PMID: 23587214
- Richer J, Hill HL, Wang Y, Yang ML, Hunker KL, Lane J, et al. A Novel Recurrent *COL5A1* Genetic Variant Is Associated With a Dysplasia-Associated Arterial Disease Exhibiting Dissections and Fibromuscular Dysplasia. Arterioscler Thromb Vasc Biol. 2020; 40(11):2686–99. https://doi.org/10.1161/ ATVBAHA.119.313885 PMID: 32938213

- Kai AK, Lam AK, Chen Y, Tai AC, Zhang X, Lai AK, et al. Exchange protein activated by cAMP 1 (*Epac1*)-deficient mice develop beta-cell dysfunction and metabolic syndrome. FASEB J. 2013; 27 (10):4122–35. https://doi.org/10.1096/fj.13-230433 PMID: 23825225
- Rim HS, Kim MG, Park DC, Kim SS, Kang DW, Kim SH, et al. Association of Metabolic Syndrome with Sensorineural Hearing Loss. J Clin Med. 2021; 10(21). <u>https://doi.org/10.3390/jcm10214866</u> PMID: 34768385
- Sun F, Zhang J, Chen L, Yuan Y, Guo X, Dong L, et al. Epac1 Signaling Pathway Mediates the Damage and Apoptosis of Inner Ear Hair Cells after Noise Exposure in a Rat Model. Neuroscience. 2021; 465:116–27. https://doi.org/10.1016/j.neuroscience.2021.03.032 PMID: 33838290
- Freeman H, Shimomura K, Horner E, Cox RD, Ashcroft FM. Nicotinamide nucleotide transhydrogenase: a key role in insulin secretion. Cell Metab. 2006; 3(1):35–45. <u>https://doi.org/10.1016/j.cmet.2005</u>. 10.008 PMID: 16399503
- Samocha-Bonet D, Wu B, Ryugo DK. Diabetes mellitus and hearing loss: A review. Ageing Res Rev. 2021; 71:101423. https://doi.org/10.1016/j.arr.2021.101423 PMID: 34384902
- Koscielny G, Yaikhom G, Iyer V, Meehan TF, Morgan H, Atienza-Herrero J, et al. The International Mouse Phenotyping Consortium Web Portal, a unified point of access for knockout mice and related phenotyping data. Nucleic Acids Res. 2014; 42(Database issue):D802–9. <u>https://doi.org/10.1093/nar/</u> gkt977 PMID: 24194600
- Ge T, Chen CY, Ni Y, Feng YA, Smoller JW. Polygenic prediction via Bayesian regression and continuous shrinkage priors. Nat Commun. 2019; 10(1):1776. <u>https://doi.org/10.1038/s41467-019-09718-5</u> PMID: 30992449
- Jiang L, Zheng Z, Fang H, Yang J. A generalized linear mixed model association tool for biobank-scale data. Nat Genet. 2021; 53(11):1616–21. https://doi.org/10.1038/s41588-021-00954-4 PMID: 34737426
- Martin AR, Kanai M, Kamatani Y, Okada Y, Neale BM, Daly MJ. Clinical use of current polygenic risk scores may exacerbate health disparities. Nat Genet. 2019; 51(4):584–91. <u>https://doi.org/10.1038/s41588-019-0379-x PMID: 30926966</u>
- Ji X, Kember RL, Brown CD, Bucan M. Increased burden of deleterious variants in essential genes in autism spectrum disorder. Proc Natl Acad Sci U S A. 2016; 113(52):15054–9. https://doi.org/10.1073/ pnas.1613195113 PMID: 27956632
- Trubetskoy V, Pardinas AF, Qi T, Panagiotaropoulou G, Awasthi S, Bigdeli TB, et al. Mapping genomic loci implicates genes and synaptic biology in schizophrenia. Nature. 2022; 604(7906):502–8. <u>https:// doi.org/10.1038/s41586-022-04434-5</u> PMID: 35396580
- Peterson RE, Kuchenbaecker K, Walters RK, Chen CY, Popejoy AB, Periyasamy S, et al. Genomewide Association Studies in Ancestrally Diverse Populations: Opportunities, Methods, Pitfalls, and Recommendations. Cell. 2019; 179(3):589–603. <u>https://doi.org/10.1016/j.cell.2019.08.051</u> PMID: 31607513
- Sirugo G, Williams SM, Tishkoff SA. The Missing Diversity in Human Genetic Studies. Cell. 2019; 177 (4):1080. https://doi.org/10.1016/j.cell.2019.04.032 PMID: 31051100
- Ioannidis NM, Rothstein JH, Pejaver V, Middha S, McDonnell SK, Baheti S, et al. REVEL: An Ensemble Method for Predicting the Pathogenicity of Rare Missense Variants. Am J Hum Genet. 2016; 99 (4):877–85. https://doi.org/10.1016/j.ajhg.2016.08.016 PMID: 27666373
- 1000 Genomes Project Consortium. A global reference for human genetic variation. Nature. 2015; 526 (7571):68–74. https://doi.org/10.1038/nature15393 PMID: 26432245
- 43. Northern JL. Hearing Disorders. 3rd ed. Boston: Allyn and Bacon; 1995.
- Moore CB, Wallace JR, Frase AT, Pendergrass SA, Ritchie MD. BioBin: a bioinformatics tool for automating the binning of rare variants using publicly available biological knowledge. BMC Med Genomics. 2013; 6 Suppl 2:S6. https://doi.org/10.1186/1755-8794-6-S2-S6 PMID: 23819467
- Taliun D, Harris DN, Kessler MD, Carlson J, Szpiech ZA, Torres R, et al. Sequencing of 53,831 diverse genomes from the NHLBI TOPMed Program. Nature. 2021; 590(7845):290–9. <u>https://doi.org/10.1038/</u> s41586-021-03205-y PMID: 33568819
- 46. Das S, Forer L, Schonherr S, Sidore C, Locke AE, Kwong A, et al. Next-generation genotype imputation service and methods. Nat Genet. 2016; 48(10):1284–7. https://doi.org/10.1038/ng.3656 PMID: 27571263
- Chang CC, Chow CC, Tellier LC, Vattikuti S, Purcell SM, Lee JJ. Second-generation PLINK: rising to the challenge of larger and richer datasets. Gigascience. 2015; 4:7. https://doi.org/10.1186/s13742-015-0047-8 PMID: 25722852
- Pruim RJ, Welch RP, Sanna S, Teslovich TM, Chines PS, Gliedt TP, et al. LocusZoom: regional visualization of genome-wide association scan results. Bioinformatics. 2010; 26(18):2336–7. https://doi.org/ 10.1093/bioinformatics/btq419 PMID: 20634204